Author's response to reviews

Title: Reduced Dicer expression in the cord blood of infants admitted with severe respiratory syncytial virus disease

Authors:

Christopher S Inchley (christopher.inchley@medisin.uio.no)
Tonje Sonerud (Tonje.Sonerud@ahus.no)
Hans Olav Fjaerli (Hans.Olav.Fjaerli@ahus.no)
Britt Nakstad (britt.nakstad@medisin.uio.no)

Version: 3 Date: 11 February 2011

Author's response to reviews: see over
Akershus University Hospital  
1478 Lørenskog  
Norway

To the Editor, BMC Infectious Diseases

Please find enclosed the revised manuscript “Reduced Dicer expression in the cord blood of infants admitted with severe respiratory syncytial virus disease.”

Again, we wish to thank the reviewers for their time and kind comments. We have amended the manuscript according to Dr Fabbiani’s recommendation. We have changed the IQR to a proper range instead of a single number in tables 2 and 3.

The manuscript has not been submitted or accepted for publication in any other journals. All authors have been significantly involved in the conception and design of the study, analysis and interpretation of the data, drafting and revision of the manuscript for intellectual content, and consent to the final version of the revised manuscript submitted to BMC Infectious Diseases. We have not received any other writing assistance.

Yours sincerely,

Christopher Inchley, MBChB  
Department of Pediatrics  
Akershus University Hospital  
Norway